Author's response to reviews

Title: NOURISH Nutritional OUtcomes from a Randomised Investigation of Intradialytic oral nutritional Supplements in patients receiving Haemodialysis. A pilot Randomised Controlled Trial

Authors:

Louise Jackson Mrs (louise.jackson@outlook.com)
Judith Cohen Dr (J.cohen@sheffield.ac.uk)
Benjamin Sully Mr (ben.sully88@gmail.com)
Steven Julious Dr (s.a.julious@sheffield.ac.uk)

Version: 3
Date: 1 March 2015

Author's response to reviews: see over
Dear Editors-in-Chief of ‘Pilot and Feasibility studies’,

Please find enclosed a revised manuscript for publication. I have endeavoured to address all your comments as detailed below:

1. I have included the number of people targeted in the abstract along with where the location of the unit. I have also included palatability of ONS as a feasibility outcome measure

2. I have gone on to reiterate the feasibility objectives within the methods section

3. More detail has been provided regarding the randomisation process

4. The results listed in table one have been commented upon

5. I have provided greater detail on how the 189 sample size was formulated

6. The numbers have been included alongside the percentages to help the reader contextualise

7. The discussion has been given some additional subheadings

8. The abbreviations have been written in full and the abbreviation list removed

I trust that these alterations are acceptable and look forward to hearing from you regarding publication.

Yours Faithfully

Louise Jackson